Author’s response to reviews

Title: Bisphosphonate-Induced Orbital Inflammation in a Patient on Chronic Immunosuppressive Therapy

Authors:
Isabella Herrera (iherrera0622@gmail.com)
Yong Kam (ykam@kumc.edu)
Thomas Whittaker (twhittaker@kumc.edu)
Mary Champion (mpham@kumc.edu)
Radwan Ajlan (radwanajlan@gmail.com)

Version: 2 Date: 30 Jan 2019

Author’s response to reviews:

Dear reviewers,

Thank you for your consideration and letter.

Please find attached our point-by-point response to your valuable comments:

1- Consent for Publication section: the consent section was updated as advised and the following was added: “Written informed consent was obtained from the patient for publication of this article and any accompanying images. A copy of the written consent is available for review by the Editor of this journal.”

2- The response to reviewers section in the manuscript, and the CARE checklist document were removed from this revised submission as recommended.

3- The "Discussion" and "Conclusions" sections were combined into one section entitled "Discussion and Conclusions" as advised.

4- A single, final, clean version of the manuscript was uploaded, while the figures (and additional files) remained uploaded as separate files as recommended.

Thank you again for your time and consideration.

Sincerely,
Radwan Ajlan